Author's response to reviews

Title: Awareness and knowledge of Sexually Transmitted Diseases (STDs) among school-going adolescents in Europe: A systematic review of published literature.

Authors:

Florence N Samkange-Zeeb (samkange@bips.uni-bremen.de)
Lena Spallek (lspallek@bips.uni-bremen.de)
Hajo Zeeb (zeeb@bips.uni-bremen.de)

Version: 5 Date: 20 September 2011

Author's response to reviews:

Dear Editor,

We have undertaken the following formatting changes as instructed:
- The writing of the author list has been changed accordingly
- Citation '57' has been corrected
- Citation numbers have been added to all references in tables

Minor changes:
- the title has been reformatted as suggested
- word count has been removed from abstract
- 'Introduction' has bee renamed 'Background'
- all reference citations have now been placed before the full stop
- 'References' now precedes 'Tables'
- visible vertical lines have been removed from all tables
- all text is black
- Highlighting has been removed from the text
- The title has been removed from the Figure page and included in the manuscript after the reference list.

Regards,

Florence Samkange-Zeeb